Author’s response to reviews

Title: FKBPL and its peptide derivatives inhibit cancer stem cells and breast cancer metastasis by downregulating DLL4 and Notch4

Authors:

Lana McClements (lana.mccllements@uts.edu.au)
Stephanie Annett (stephanieannett@rcsi.ie)
Anita Yakkundi (yakkundi@hotmail.com)
Martin O'Rourke (martinorourke16@gmail.com)
Andrea Valentine (andrea.valentine@crl.com)
Nermeen Moustafa (nmoustafa01@qub.ac.uk)
Abdelrahim Alqudah (aalqudah02@qub.ac.uk)
Bruno Simões (Bruno.Simoes@ics.manchester.ac.uk)
Fiona Furlong (f.furlong@qub.ac.uk)
Amy Short (ashort04@qub.ac.uk)
Stuart McIntosh (s.mcintosh@qub.ac.uk)
Helen McCarthy (h.mccarthy@qub.ac.uk)
Robert Clarke (Robert.Clarke@ics.manchester.ac.uk)
Tracy Robson (tracyrobson@rcsi.ie)

Version: 1 Date: 03 Jan 2019

Author’s response to reviews:

Editor's comments:

1 - To maintain anonymity for the participants, we have removed the gender and exact age from the supplementary table 1.

2 - We have included references in methods where there was textual overlap - to ensure that we have summarized the methods and cited the source.
3 - We have clearly stated the source of Mice in the methods.

4 - We have included an animal ethics statement in the 'Ethics approval and consent to participate' section of the Declarations.

5 - We have modified the manuscript to adhere to the ARRIVE guidelines (http://www.nc3rs.org.uk/arrive-guidelines) for the reporting of animal experiments. This is so your methodology can be fully evaluated and utilised.

6 - Please provide a heading for the Declarations. - done

7 - In the Funding section of the Declarations please indicate the role of the funding body in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript. If no specific funding was received for this study, please clearly indicate this in the Funding section. - done